Author’s response to reviews

Title: Laparoscopic repair of secondary parahiatal hernia with incarceration of the stomach: a case report

Authors:

Masashi Takemura (mtake@hyo-med.ac.jp)
Katsuyuki Mayumi (k_mayumi@me.com)
Takashi Ikebe (m0269705@msic.med.osaka-cu.ac.jp)
Genya Hamano (m2059356@med.osaka-cu.ac.jp)

Version: 2 Date: 3 December 2012

Author’s response to reviews:

Dear Editor in Chief of Journal of Medical Case Reports

MS: 1924842551820041
Title: Laparoscopic repair of secondary parahiatal hernia with incarceration of the stomach: a case report
Masashi Takemura, Katsuyuki Mayumi, Takashi Ikebe and Genya Hamano

Thank you for your kind letter for our manuscript. I am sending herewith the revised version of our manuscript.

Our comments for the reviewers’ suggestion and The Editorial Board are follow as below.

Masashi Takemura, MD

Commnets for reviewew 1
1. We replaced the “heavy suture” to “0 PDS II”.
2. We added the sentence and references of the effect of the chronic intraabdominal pressure for creation of diaphragmatic hernia (page 7, line 11-21).

Commnets for reviewew 2
1. The long-term hospitalization of our case caused by a wish of a patient.
2. We added the sentence and references of the effect of the chronic intraabdominal pressure for creation of diaphragmatic hernia (page 7, line 11-21). Moreover, we added the “iatrogenic” in discussion.

Commnets for Editor
1. We removed the running title.
2. We include the ethnicity of the patient in the abstract.
3. We replace to the “Case presentation”.
4. We replace to the “male”.
5. We replace to the “Consent”.
6. We revise the consent form from the patient.